Author's response to reviews

Title: Effect of Fagonia Arabica (Dhamasa) on in vitro thrombolysis

Authors:

Sweta Prasad (sweta_ciims@yahoo.com)
Rajpal S Kashyap (raj_ciims@rediffmail.com)
Jayant Y Deopujari (jy_deopujari@rediffmail.com)
Hemant J Purohit (hemantdrd@rediffmail.com)
Girdhar M Taori (taorigm_ciims@yahoo.co.in)
Hatim F Daginawala (hfd_ciims@rediffmail.com)

Version: 5 Date: 3 November 2007

Author's response to reviews:

To,
The BioMed Central Editorial Production Team

Dear Sir/Madam,

Please find enclosed copy of the revised manuscript entitled Effect of Fagonia Arabica (Dhamasa) on in vitro thrombolysis (MS:1588256974135811). We have gone through the manuscript thoroughly. Formatting changes mentioned in the mail has been done and are written point-to-point below.

With best regards,

Dr. H.F. Daginawala

Formatting Changes done are:

1. Full stop has been removed from the end of the title.
2. Red text changed to black.
3. Materials and Methods section has been renamed as “Methods”.
4. Declaration of competing interest has been changed to “Competing interests”.
5. In competing interest section author(s) has been changed to author.
6. Author’s contribution has been changed to authors’ contribution.
7. Table title and the legend are now above and below the table respectively.
8. All blank pages has been deleted.
9. Figures are cropped closely to minimize white space around the image.